Title: Array-CGH in patients with Kabuki-like phenotype: Identification of two patients with complex rearrangements including 2q37 deletions and no other recurrent aberration

Authors:

Ivon Cuscó (ivon.cusco@upf.edu)
Miguel Del Campo (miguel.delcampo@upf.edu)
Mireia Vilardell (mvilardell@prbb.org)
Eva González (eva.gonzalez@crg.es)
Blanca Gener (BLANCA.GENERQUEROL@osakidetza.net)
Enrique Galán (egalan@unex.es)
Laura Toledo (hdezmarti@inicia.com)
Luis A Pérez-Jurado (luis.perez@upf.edu)

Version: 5 Date: 8 April 2008

Author's response to reviews: see over
Dear Editors:

Please find enclosed our revised manuscript with all the formatting changes requested.

We have made all the changes suggested:

1) Authors list. We have modified the authors name in the submission system, and now are written exactly.

2) Website addresses. We have removed the websites from the text and they are cited in the main reference list only.

3) Tables’ size. The two tables are now into an A4 portrait page (one in the horizontal format to fit)

4) Additional files. All additional files are mentioned in the text, and we have added a new section (called Additional files) at the end of the manuscript (after the tables) with all the information of these additional files.

We hope this revised manuscript is acceptable for publication in its present form.

Sincerely yours,

Corresponding author: Luis A. Pérez Jurado, MD, PhD.
Associate Professor of Genetics
Genetics Unit, Department of Experimental Sciences, Universitat Pompeu Fabra
Dr. Aiguader 88
08003, Barcelona, Spain
Phone: +34 93 316 0856; Fax: +34 93 316 0901
e-mail: luis.perez@upf.edu